Author’s response to reviews

Title: Addressing malnutrition among children in routine care: How is the Integrated Management of Childhood Illnesses strategy implemented at health centre level in Burundi?

Authors:

Manassé NIMPAGARITSE (mnimpagaritse@yahoo.fr)

Catherine Korachais (ckorachais@itg.be)

Georges Nsengiyumva (georges.nsengiyumva@gmail.com)

Jean Macq (jean.macq@uclouvain.be)

Bruno Meessen (bmeessen@itg.be)

Version: 2 Date: 17 Nov 2018

Author’s response to reviews:

1. Please include the date of trial registration with the Trial registration number in the abstract. If the date of registration is after the date that the first participant was recruited, please also state ‘retrospectively registered’.

   □ Author: Addressed as suggested (line 55)

   Added: March 2016 (retrospectively registered)

2. In your Funding section, please also state the role of the funding body in the design of the study; collection, analysis, and interpretation of data; and in writing the manuscript.

   □ Author: Addressed as suggested (line393-394 & line 397-398)

3. Submission of a manuscript to a BioMed Central journal implies that materials described in the manuscript, including all relevant raw data, will be freely available to any scientist wishing to use them for non-commercial purposes, without breaching participant confidentiality. Thus, we request that you upload English language versions of all questionnaires developed specifically for use in this study as supplementary files. If the
questionnaires are not yours or are already published by you, please supply references and/or links to them.

Author: The questionnaires used for this study were developed for a larger impact evaluation of the performance based financing (PBF) scheme applied to malnutrition in Burundi’s Health Centres. They were published elsewhere as well as other tools used in the baseline. (line 370-376 & line 549-555)

4. Please provide more justification for the contributions of authors. Currently, the contributions of Georges Nsengiyumva do not automatically qualify for authorship. Please amend your manuscript accordingly.

An 'author' is generally considered to be someone who has made substantive intellectual contributions to a published study. We recommend that that you adhere to the guidelines for authorship that are applicable in your research field or, in the absence of any guidelines, to the International Committee of Medical Journal Editors (ICMJE) guidelines. According to the ICMJE guidelines, to qualify as an author one should have:

- made substantial contributions to conception and design, or acquisition of data, or analysis and interpretation of data;

- been involved in drafting the manuscript or revising it critically for important intellectual content;

- given final approval of the version to be published. Each author should have participated sufficiently in the work to take public responsibility for appropriate portions of the content; and

- agreed to be accountable for all aspects of the work in ensuring that questions related to the accuracy or integrity of any part of the work are appropriately investigated and resolved.
Making comments, alone, does not usually justify authorship. Please clarify if the comments provided were critical to the revision and final form of the manuscript, or please detail any additional contributions GN made to the study/manuscript which may warrant authorship.

Please consider the list of authors as it currently stands with reference to our guidelines regarding qualification for authorship (http://www.biomedcentral.com/submissions/editorial-policies#authorship).

- Author: Sorry for not being precise, addressed (line 400-408)

5. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

- Author: Addressed as suggested